Author's response to reviews

Title: The effect of tracheotomy on drug consumption in patients with acute aneurysmal subarachnoid hemorrhage: an observational study.

Authors:

Leiv A Rosseland (l.a.rosseland@medisin.uio.no)
Jon Narum (jnarum@ous-hf.no)
Audun Stubhaug (audun.stubhaug@medisin.uio.no)
Ulf Kongsgaard (ulf.kongsgaard@medisin.uio.no)
Wilhelm Sorteberg (wilhelm.sorteberg@ous-hf.no)
Angelika Sorteberg (angelika.sorteberg@ous-hf.no)

Version: 6  Date: 10 February 2015

Author's response to reviews:

Thank you again for addressing important questions.
The abstract has been changed and identicality has been secured.
Ventilatory modes are explained in the text reporting the results, as suggested.
Regarding inter-quartile range this is in my textbook in Medical statistics (Altman et al) defined as the numerical difference between the 25th and the 75th percentile. Hence, this is a number, not a range. We have not made any changes in this regard. Is this acceptable?

Best regards,
Leiv Arne Rosseland